Author's response to reviews

Title: Stay@Work: Participatory Ergonomics to prevent low back and neck pain among workers: design of a randomised controlled trial to evaluate the (cost-)effectiveness

Authors:

Maurice T Driessen (m.driessen@vumc.nl)
Johannes R Anema (h.anema@vumc.nl)
Karin I Proper (k.i.proper@vumc.nl)
Paulien M Bongers (paulien.bongers@tno.nl)
Allard J van der Beek (a.vanderbeek@vumc.nl)

Version: 4 Date: 19 September 2008

Author's response to reviews:

Dear editor,

In our revised manuscript we made 2 minor changes. We adjusted the number of participants in figure 1 and changed n=2245 to n=1384 which is the right number of workers needed for company 3 and 4. Furthermore, some changes were made to the reference list. For used publications, we only wrote down 5 or less authors and abbreviated the author list by using et al. However, other biomedical central authors wrote down all authors, therefore we changed this in our manuscript and wrote down all authors. We hope that the document can be accepted in this form.

On behalf of all authors,

Maurice Driessen